Author's response to reviews

Title: Screening for atrial fibrillation with baseline and intermittent ECG recording in an out-of-hospital population

Authors:

Tijn Hendrikx (tijn.hendrikx@fammed.umu.se)
Rolf Hörnsten (rolf.hornsten@clinphys.umu.se)
Mårten Rosenqvist (marten.rosenqvist@sodersjukhuset.se)
Herbert Sandström (herbert.sandstrom@fammed.umu.se)

Version: 6 Date: 22 May 2013

Author's response to reviews: see over
Thank you very much for giving the opportunity to address the reviewer’s comments in a revised manuscript. The following cover letter will give a point-to-point response to the concerns.

**Reviewer:** Marco Perez, **date** 15 May 2013

One discretionary suggestion is to change the title to "Screening for atrial fibrillation with baseline and intermittent ECG recording in an out-of-hospital population" to make clearer the fact that baseline recordings were also used to catch several episodes of AF.

*The title of the manuscript is changed to "Screening for atrial fibrillation with baseline and intermittent ECG recording in an out-of-hospital population".*

T. Hendriks, MD, Family Medicine, Department of Public Health and Clinical Medicine, Umeå University, SE-901 87 Umeå.
E-mail: tijn.hendriks@fammed.umu.se

Rolf Hörnsten PhD
M. Rosenqvist MD PhD
H. Sandström MD PhD